Author’s response to reviews

Title: The effects of metformin in type 1 diabetes mellitus

Authors:

Selvihan Beysel (beyselsevilvan@gmail.com)
Ilknur Unsal (ilknur_dr@yahoo.com)
Muhammed Kizilgul (muhammedkzgl@gmail.com)
Mustafa Caliskan (mcalisikan37@yahoo.com)
Bekir Ucan (uzm.dr.bekir@hotmail.com)
Erman Cakal (ermancakal@hotmail.com)

Version: 4 Date: 20 Nov 2017

Author’s response to reviews:

1- 'Availability of data and material: The datasets used and/or analysed during the current study available from the corresponding author on reasonable request'. This sentence was added to declarations

2- 'Ethics approval and consent to participate: Approval of ethical committee and written informed consent of participants were obtained before the study'. This sentence was added to ethics and consent to participate part

3- Our manuscript was formatted according to guidelines

4- English editing was done